\documentclass[11pt]{article}

%-----------------------------------------------------------------------
%
% definitions, macros, etc.
%

\def\CourseNum{CSCI 2194}
\def\CourseName{Professional, Ethics, and Design Seminar}
\def\Term{Spring 2011}
\def\DocTitle{Final Group Evaluation}

%\input{my-macros-for-class}
% standard bunch of packages
\usepackage{calc}
\usepackage{moreverb}
\usepackage{array}
\usepackage{latexsym}
\usepackage{verbatim}
\usepackage{tabularx}
\usepackage{graphicx}
\usepackage{html}
\usepackage{url}
\usepackage[normalem]{ulem}	% has \uline
\usepackage{mdwlist}		% has \itemize*

\usepackage[english]{babel}
\selectlanguage{english}
%
% page setup
%
\usepackage{vmargin}
\setpapersize{USletter}
\setlength{\headsep}{4ex}
\def\MySideMargin{1in}
\def\MyTopMargin{0.5in}
\setmargrb{\MySideMargin}{\MyTopMargin}{\MySideMargin}{\MyTopMargin}
%
% headings
%
\usepackage{fancyheadings}
\lhead{{\bf \CourseNum}}
\chead{{\bf \DocTitle}}
\rhead{{\bf \Term}}
\pagestyle{fancy}
%
% allow more figures on a page
%
\setcounter{topnumber}{10}
\setcounter{bottomnumber}{10}
\setcounter{totalnumber}{10}
\setcounter{dbltopnumber}{10}
\def\topfraction{1.0}
\def\bottomfraction{1.0}
\def\textfraction{0.0}
\def\dbltopfraction{1.0}

%-----------------------------------------------------------------------

\title{\CourseNum\ (\CourseName), \Term:  \DocTitle}
\author{}
\date{}

\begin{document}

%-----------------------------------------------------------------------
%
% header
%

\thispagestyle{empty}

\begin{latexonly}
	\begin{center}
	\begin{Large}
	\textbf{\CourseNum\ (\CourseName), \Term} \\ \bigskip
	\textbf{\DocTitle}
	\end{Large}
	\end{center}
	\bigskip
\end{latexonly}

\begin{htmlonly}
%\maketitle
\HTMLcode{H1}{\CourseNum\ (\CourseName), \Term: \\ 
	\DocTitle}
\end{htmlonly}

% FIX THIS?
%\tableofcontents

\vspace*{2ex}

\begin{large}
\noindent
{\bf Name and Group Number:} \hrulefill \\
\end{large}

\vspace*{6ex}

%-----------------------------------------------------------------------
%
% body
%

\noindent
Describe the work you did on this part of the project
(design of implementation, development of prototype).

\vfill

\noindent
How well do you think your group did with this part of the
project?  What did you do well?  What did you not do well?
What if anything would you do differently if you were
starting over?

\vfill

\noindent
Evaluate the contributions of the other members of your group.
For each other member, did he or she attend group meetings
and participate?  contribute to the project?  work well with
other group members?  Give each person a numerical rating
(with 1 the best rating and 5 the worst) and comment if you
like.  
	
\begin{center}
\begin{tabular}{|p{0.2\linewidth}|p{0.1\linewidth}|p{0.60\linewidth}|}
\hline
Name & Rating & Comments \\
\hline
 & & \\
 & & \\
 & & \\
\hline
 & & \\
 & & \\
 & & \\
\hline
 & & \\
 & & \\
 & & \\
\hline
 & & \\
 & & \\
 & & \\
\hline
\end{tabular}
\end{center}

\end{document}

